Author's response to reviews

Title: Can international health programmes be sustained after the end of international funding? The case of eye care interventions in Ghana

Authors:

  Karl Blanchet (karl.blanchet@lshtm.ac.uk)
  Philip James (P.James@salford.ac.uk)

Version: 4 Date: 5 August 2013

Author's response to reviews:

Dear reviewers,

thank you very much for your valuable comments.

I have uploaded a new version of the paper based on the comments from both reviewers.

Financing: I have added a full section on financing explaining how the financial situation changed after the end of international funding.

Methodology:
- I have provided further details on the qualitative methodology explaining how data collected through interviews helped analyse the scoring.
- More details have been added in relation to the year of data collection and analysis, which took place in 2009.
- The validity of the approach is discussed in relation to the methods used.

Limitation: Two sections on limitations have been added: one in the methodology chapter and a second one at the end of the discussion chapter.

- The word "developing country" has been replaced by low income country.
- I have maintained the tables 7-12 as I haven't found any alternative on how to present the information.

I look forward to hearing from you.

Best regards